Author’s response to reviews

Title: Larger Omental Adipocytes Correlate with Greater Fetuin-A Reduction Following Sleeve Gastrectomy

Authors:
Katie Robinson (katie-robinson@uiowa.edu)
Blair Rowitz (Blair.Rowitz@carle.com)
Uretz Oliphant (Uretz.Oliphant@carle.com)
Sharon Donovan (sdonovan@illinois.edu)
Margarita Teran-Garcia (teranmd@illinois.edu)

Version: 2 Date: 04 Mar 2019

Author’s response to reviews:

REVISION RESPONSE LETTER

Thank you very much for your feedback. The revisions are detailed below.

--Please include your title page within your manuscript, rather than as a supplementary information file.

The title page has been added to the manuscript.

-- Figure files should contain only the image/graphic, as well as any associated keys/annotations. If titles/legends are present within the figure files, please remove them.

Titles have been removed from all figure files.
-- Please provide figure titles/legends under a separate heading of 'Figure Legends' after the References. If Figure titles/legends are within the main text of the manuscript, please move them.

Figure titles/legends are listed under a separate heading of 'Figure Legends' after the References

--Please confirm why Dr. Teran-Garcia reports grants from USDA National Institute of Food and Agriculture as a conflict of Interest?

Thank you for catching this error. Dr. Teran’s COI form has been updated to reflect no conflicts of interest.

--All financial and non-financial competing interests must be declared in the 'Competing interests' section.

If you do not have any competing interests, please state "The authors declare that they have no competing interests" in this section.

The competing interest statement is found on page 17, line 12

-- Please ensure that all figures/tables and supplementary files are cited within the text. Any items which are not cited may be deleted by our production department upon publication. (Fig 6)

Figure 6 was referenced on page 12, lines 2-5.

-- Thank you for providing a response to the reviewers/Cover Letter. As these documents are no longer required at this stage of the publication process, please remove them from your submission’s supplementary files.
The “response to reviewers” documented has been removed from the submission.

-- At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Clean versions of the manuscript and figures 1-6 have been added to the current submission.

We look forward to receiving your feedback and please do not hesitate to contact us if you have any questions.

Best wishes,
Katie Robinson and Margarita Teran-Garcia